Author's response to reviews

Title: Gatekeepers of health: a qualitative assessment of child care centre staff's perspectives, practices and challenges to enteric illness prevention and management in child care centres

Authors:

Marsha Taylor (marsha_taylor@phac-aspc.gc.ca)
Cindy L Adams (cadams@ucalgary.ca)
Andrea Ellis (andrea_ellis@phac-aspc.gc.ca)

Version: 4 Date: 22 May 2008

Author's response to reviews: see over
May 21, 2008

Dear Dr. da-Silva:

Attached please find a revised copy of the manuscript entitled, Gatekeepers of health: A qualitative assessment of child care centre staff’s perspectives, practices and challenges to enteric illness prevention and management in child care centres. We are very happy that BMC-Public Health has chosen to publish this research article.

All changes to format as suggested by the Editorial Production Team have been made and include: removing the headers, Authors and Affiliations as well as author qualifications from the title page and ensuring all blank pages have been removed. At this time we feel that the manuscript is ready for publication.

We appreciate your consideration and acceptance of the manuscript for publication as well as the work of the reviewers, editorial and production teams.

Sincerely,

Ms. Marsha Taylor
Epidemiologist
Public Health Agency of Canada
marsha_taylor@phac-aspc.gc.ca